Author's response to reviews

Title: Equal antipyretic effectiveness of oral and rectal acetaminophen: a randomized controlled trial [ISRCTN11886401]

Authors:

Mona Nabulsi (mn04@aub.edu.lb)
Hala Tamim (ht02@aub.edu.lb)
Ramzi Sabra (rsabra@aub.edu.lb)
Ziyad Mahfoud (zm15@aub.edu.lb)
Shadi Malaeb (smalaeb@wihri.org)
Hadi Fakih (alfakihdr@hotmail.com)
Mohammad Mikati (mamikati@aub.edu.lb)

Version: 7 Date: 14 August 2005

Author's response to reviews: see over
Re: "Equal antipyretic effectiveness of oral and rectal acetaminophen: a randomized controlled trial [ISRCTN11886401]"

Dear Editor,

Thank you for accepting our manuscript. We have made the following changes in the manuscript as per your suggestion:

1. The list of authors in the manuscript is now written exactly as those in the submission system.
2. Page breaks have been removed.
3. Keywords section has been deleted.
4. Consort checklist has been removed from "Additional files".
5. The manuscript has been checked for typing errors and is now approved by all co-authors as the final version.

Thank you very much and best regards.

Sincerely,

Mona Nabulsi, MD
Associate Professor of Pediatrics
Department of Pediatrics
American University of Beirut Medical center
Beirut-Lebanon